Author's response to reviews

Title: Antioxidants and breast cancer risk - a population-based case-control study in Canada

Authors:

Sai Yi Pan (sai.yi.pan@phac-aspc.gc.ca)
Jia Zhou (zhoujia10@163.com)
Laurie Gibbons (laurie.gibbons@phac-aspc.gc.ca)
Howard Morrison (howard.morrison@phac-aspc.gc.ca)
Shi Wu Wen (sswen@ohri.ca)

Version: 6 Date: 18 August 2011

Author's response to reviews:

As the original agreement of data collection, the Canadian Cancer Registries Epidemiology Research Group has to be included as a coauthor.